Author's response to reviews

Title: Exenatide Once Weekly Treatment Maintained Improvements in Glycemic Control and Weight Loss Over 2 Years

Authors:

Kristin Taylor (kristin_taylor2000@yahoo.com)  
Kate Gurney (kate.gurney@amylin.com)  
Jenny Han (j.han@amylin.com)  
Richard Pencek (richard.pencek@amylin.com)  
Brandon Walsh (brandon.walsh@amylin.com)  
Michael Trautmann (trautmann_michael@lilly.com)

Version: 7 Date: 20 April 2011

Author's response to reviews:

20 April, 2011
Re: MS 1343862011464658

Dear Dr. Shipley,

The proofs of the accepted manuscript “Exenatide Once Weekly Treatment Maintained Improvements in Glycemic Control and Weight Loss Over 2 Years” have been checked and minor corrections have been made. The manuscript is now ready for publication.

Please contact me with any additional questions or issues.

Sincerely,
Kate Gurney

9360 Towne Centre Drive
San Diego, CA 92121 USA
Telephone: 858-458-8520
kate.gurney@amylin.com